Editorials

Breaking the cycle of inequalities for people with a learning disability

A woman with a learning disability is likely to die from similar causes to the general population, but on average 20 years earlier than a woman without a learning disability; a man with a learning disability is likely to die on average 13 years earlier. A learning disability is not a life-limiting illness and yet it continues to be associated with significant health inequalities.

During the pandemic, adults aged 18-34 years with a learning disability were 30 times more likely to die of COVID-19 than young adults in the general population.2 Initial National Institute for Health and Care Excellence quidance (NG159) categorised many people with a learning disability into frailty categories, lowering their prioritisation for emergency triage, a situation exacerbated by the use of 'do not resuscitate' notices, sometimes without consultation.3 It wasn't until February 2021 that people with a mild or moderate learning disability were placed in the vaccination priority groups: until then, carers of people with a learning disability were being invited for vaccine before those that they cared for.

Experience of this pandemic-related inequality and injustice has broken the fragile trust many people with a learning disability and their families had with the

Change is coming: a new digital flag for reasonable adjustments on all summary care records (currently piloting) and the roll-out of the Oliver McGowan mandatory training on learning disability and autism for all health and social care staff.4 The Department of Health and Social Care will be publishing a Code of Practice for the training next year. In anticipation, there are changes that can be made now, in clinical practice and in the way we conduct our research.

CHANGES TO CLINICAL PRACTICE

I will take diabetes as an exemplar. Primary

"Experience of this pandemic-related inequality and injustice has broken the fragile trust many people with a learning disability and their families had with the NHS."

care management of diabetes for people with a learning disability is not preventing emergency admissions for diabetic complications,⁵ for several reasons. The first is learning disability registers. Coding is variable in primary care.6 It is suggested that only 25% of people eligible to be on the register are on it.7 In one clinical commissioning group we found 16 people whose records contained coding for Down syndrome yet they were not on the learning disability register.6 The code 918E is recommended and we should be using it now. It can ensure that the individual will be invited for a learning disability health check, diabetes care plan, and any relevant vaccine

LEARNING FROM A RANDOMISED CONTROLLED TRIAL

During a randomised controlled trial on the management of type 2 diabetes we spoke to 172 people with a learning disability and diabetes in West Yorkshire. Twenty-five were not eligible for the study. For nearly half this was because their diabetes was not type 2; the people caring for them had referred them because they didn't know the difference.8 This highlighted a significant need for better understanding of diabetes in paid and family supporters.

We interviewed every person in that trial about their learning disability and about their ability to read. Not one person took offence. These conversations need to happen in primary care to establish people's communication needs and the outcome must be recorded.9 Many people told us that they couldn't read and saved up letters with NHS logos on them until a time when they could get support to read them, sometimes missing appointments as a result (Amy M Russell, unpublished data, 2018). Therefore, when people miss appointments, consider if you have communicated appropriately about their appointment: is your letter in a pile waiting to be read?

Over half of the participants had a mobile phone, but very few had a smart phone. Only 22% used the internet, a third of whom needed help to do so.8 Any purely webbased service can perpetuate inequalities in adults with a learning disability.

Our trial found that glycaemic control in people with a learning disability was comparable with the general population of people with type 2 diabetes, but there were high levels of obesity and low levels of physical activity in this group. When planning activity interventions, we found that high levels of fear and experiences of hate crime inhibited people's capacity to exercise in public close to home.8 Safety and transport therefore require discussion in any conversation about physical activity so that unwarranted assumptions are not made. All activity-based interventions need a discussion of wider social and economic barriers.¹⁰

WHERE TO START MAKING REASONABLE **ADJUSTMENTS IN CLINICAL PRACTICE**

General practices were willing to change but didn't always know where to start to improve their care. We created a suite of resources to support clinicians, available on the Diabetes UK website,11 including a co-produced easy-read guide on how to support informed diabetes management.12 The site is updated with new resources as we develop them.

We evaluated these resources. We found that clinicians believed they needed specialist training to work with people with a learning disability, and that coding and making

"Current research suggests people with a learning disability are excluded from 90% of health research automatically [...] small changes in clinical practice and research design can make a big difference to people with a learning disability."

reasonable adjustments were perceived as someone else's role. We need to dispel these myths; addressing this inequality is everyone's business, and special training is rarely needed for general support.

A scoping review of access to health care and public health interventions for persons with a disability found evidence of disabling attitudes towards people with disabilities within clinicians, for example, the belief that people with a disability weren't suitable for certain services, wouldn't engage with certain treatments, or didn't have a quality of life worth investing resources in. 13 Decisions were often made without consulting with the person or their family. Data evaluating the effects of disability on healthcare interventions are also lacking and we are calling for all researchers and clinicians to record disability, and particularly learning disability as standard, to improve the quality of these data.

CHANGES TO RESEARCH PRACTICES

Current research suggests people with a learning disability are excluded from 90% of health research automatically. 14 Yet there is an increased and timely interest in community engagement and co-production. To capitalise on this zeitgeist, we must place greater value on stakeholder engagement, recognising it as a research skill, and sustainably employ researchers in a way that allows them to build enduring relationships with community groups, including groups of people with learning disabilities. We must cost for, and

pay, these community groups properly for contributions throughout the research cycle from grant development to dissemination. We must design inclusivity into our research: in our materials, processes, and recruitment strategies. There are resource implications, but now is the time to ask for this funding. The door is open to such requests in ways it hasn't been before.

We need to go beyond blanket exclusions about people who lack capacity and investigate what people need to enable their informed consent. Crucially, we need to support researchers in the field who often struggle to implement capacity assessments and often, therefore, err on the side of caution. These actions will help address inequalities through increased involvement of historically excluded groups.

CONCLUSION

Inequalities in health for people with a learning disability raise questions about our systems of care. Inequalities in our evidence base raise questions about the inclusivity of our systems of research design, funding, and approval; systems we must begin to question. While that conversation takes place my message is simple: small changes in clinical practice and research design can make a big difference to people with a learning disability.

Amy M Russell,

(ORCID: 0000-0002-8891-9059) Senior Research Fellow, Leeds Institute of Health Sciences, University of Leeds, Leeds.

ADDRESS FOR CORRESPONDENCE

Amv M Russell

Level 10 Worsley Building, Clarendon Way, University of Leeds, Leeds LS2 9NL, UK.

Email: a.m.russell@leeds.ac.uk @Amy_M_Russell

Funding

This article reports findings from the National Institute for Health and Care Research (NIHR) Health Technology Assessment (reference: 10/102/03). The views expressed are those of the author and not necessarily those of the NIHR or the Department of Health and Social Care. This article also reports findings from studies funded by the Wellcome Trust (reference: 219754/Z/19/Z), British Medical Association (BMA) Scholarship Grant, Leeds Community Healthcare Trust, Leeds and York Partnership Foundation Trust, and a review conducted with colleagues at the Disability Team at the World Health Organization (WHO) (forthcoming 2022). The views expressed in this publication are those of the author and not necessarily those of the BMA, Leeds and York Partnership Foundation Trust, Leeds Community Trust, or the WHO.

Provenance

Commissioned; not externally peer reviewed.

Competing interests

Amy M Russell is supported by a Wellcome Trust grant (reference: 219754/Z/19/Z).

Acknowledgements

The author would like to thank the University of Leeds colleagues who worked with them on the research cited here and who are named in the references, especially Allan House and Louise Bryant.

DOI: https://doi.org/10.3399/bjgp22X720953

REFERENCES

- 1. Heslop P, Blair PS, Fleming P, et al. The Confidential Inquiry into premature deaths of people with intellectual disabilities in the UK: a population-based study. Lancet 2014; **383(9920):** 889-895.
- 2. Public Health England. Deaths of people identified as having learning disabilities with COVID-19 in England in the spring of 2020. 2020. https://assets.publishing.service.gov. uk/government/uploads/system/uploads/ attachment data/file/933612/COVID-19 learning_disabilities_mortality_report.pdf (accessed 12 Sep 2022).
- 3. Heslop P, Byrne V, Calkin R, et al. Deaths of people with intellectual disabilities: analysis of deaths in England from COVID-19 and other causes. J Appl Res Intellect Disabil 2021; 34(6): 1630-1640.
- Marriott A, Cooper M. Evaluation of the Oliver McGowan mandatory training in learning disability and autism. https://www.ndti.org.uk/ projects/evaluation-of-the-oliver-mcgowanmandatory-training-in-learning-disability-andautism (accessed 12 Sep 2022).

- 5. Glover G, Evison F. Hospital admissions that should not happen: admissions for ambulatory care sensitive conditions for people with learning disabilities in England. 2013. https:// www.ndti.org.uk/assets/files/IHAL-2013-02_Hospital_admissions_that_should_not_ happen_ii.pdf (accessed 12 Sep 2022).
- Russell AM, Bryant L, House A. Identifying people with a learning disability: an advanced search for general practice. Br J Gen Pract 2017; DOI: https://doi.org/10.3399/ bjgp17X693461.
- Emerson E, Glover G. The 'transition cliff' in the administrative prevalence of learning disabilities in England. Tizard Learning Disability Review 2012; 17(3): 139-143.
- 8. House A, Bryant L, Russell AM, et al. Managing with learning disability and diabetes: OK-Diabetes — a case finding study and feasibility randomised controlled trial. Health Technol Assess 2018; 22(26): 1-328.
- NHS England. Accessible information standard. 2015. https://www.england.nhs. uk/about/equality/equality-hub/patientequalities-programme/equality-frameworks-

- and-information-standards/accessibleinfo (accessed 13 Sep 2022).
- 10. House A, Latchford G, Russell AM, et al. Development of a supported selfmanagement intervention for adults with type 2 diabetes and a learning disability. Pilot Feasibility Stud 2018; 4: 106.
- 11. Diabetes UK. Improving care for people with diabetes and a learning disability. https:// www.diabetes.org.uk/professionals/resources/ shared-practice/for-people-with-learningdisability (accessed 12 Sep 2022).
- 12. Russell AM, House A, Byant L, Rabbee J. What do I need to know about diabetes? 2022. https://cdn.shopify.com/s/files/1/1922/6045/ files/easy-read-what-do-i-need-to-knowabout-diabetes.pdf (accessed 12 Sep 2022).
- 13. World Health Organization. WHO global report on health equity for persons with disabilities. Geneva: WHO, in press.
- 14. Feldman MA, Bosett J, Collet C, Burnham-Riosa P. Where are persons with intellectual disabilities in medical research? A survey of published clinical trials. J Intellect Disabil Res 2014; 58(9): 800-809.